Author's response to reviews

Title: (Cost)-effectiveness of case-management by district nurses among primary informal caregivers of older adults with dementia symptoms and the older adults who receive informal care: design of a randomized controlled trial [ISCRN83135728]

Authors:

Aaltje PD Jansen (d.jansen@vumc.nl)
Hein PJ van Hout (hpj.vanhout@vumc.nl)
Harm WJ van Marwijk (hwj.vanmarwijk@vumc.nl)
Giel Nijpels (g.nijpels@vumc.nl)
Martine C de Bruijne (mc.debruyne@vumc.nl)
Judith E Bosmans (je.bosmans@vumc.nl)
Anne-Margriet Pot (am.pot@vumc.nl)
Wim AB Stalman (w.stalman@vumc.nl)

Version: 4 Date: 26 November 2005

Author's response to reviews:

-at page 1: all zipcodes in Amsterdam are now identical: 1081 HV
-at page 3, at the top: Over the next years the number of demented older persons will increase (in stead of demented older adults)
-at page 8, in the middle: (MDS) that assesses general functioning (in stead of , which assesses..)
-at page 11, at the top: (SPICC)[50] was changed to (SPICC) [50].
-at page 16, at the top: Possibly, they will refuse more often...was changed to Possibly, such caregivers will refuse more often.
-at page 16, at the bottom: EQ-5D between patients' and caregivers' was changed to 'EQ-5D between patients and caregivers.